Author's response to reviews

Title: Determinants of sexual health knowledge among adolescent girls in schools of Riyadh city, Saudi Arabia

Authors:

Johara Al quaiz (jalquaiz@yahoo.com)
Ambreen kazi (akamran@ksu.edu.sa)
Maha Al Muneeef (mahamuneeef@gmail.com)

Version: 2 Date: 8 November 2012

Author's response to reviews: see over
To
The Editor
Date: 8th November, 2012
BMC Women’s Health

Submission of article titled “Determinants of sexual health knowledge among adolescent girls in schools of Riyadh city, Saudi Arabia”

Dear Editor,
I am resubmitting enclosed research article entitled “Determinants of sexual health knowledge among adolescent girls in schools of Riyadh city, Saudi Arabia”, for review and publishing in the BMC Women’s Health.

I have read the current Instructions to Authors and I along with co-authors comply with those instructions and acceptance of the conditions posed.

I have answered the following queries/comments made:
1. The consent was taken from the school administration as the school authorities are entitled to give the consent for non clinical studies on behalf of parents.
2. The manuscript has been formatted according the Journals standard format style.
3. Heading related to “Competing interests” added
4. Authors contribution added according to the “Instruction to authors” guide.

This paper is based on determinants of sexual health knowledge among adolescent girls in Riyadh, Saudi Arabia. The role of women in Saudi Arabia is undergoing transitional phase. Studies like the current one if published in BMC Women Health shall inform the readers world over regarding the important factors affecting the adolescent girls health in Saudi Arabia and will also help the health workers, social mobilizers, policy makers to make and implement affective programs according to the needs.

There is no perceived or real conflict of interest. The study was self sponsored and data was collected with the help of undergrad medical students.

Looking forward to a favorable response.

Sincerely,

Dr. Ambreen Kazi
Assistant Professor
Princess Nora Bint Abdullah Women Health Research Chair
College of Medicine, King Saud University, Riyadh, Saudi Arabia